Author's response to reviews

Title: A prospective randomised trial comparing nasogastric with intravenous hydration in children with bronchiolitis (protocol) The comparative rehydration in bronchiolitis study (CRIB)

Authors:

Paediatric Research in Emergency Departments International Colla (PREDICT) (ed.oakley@southernhealth.org.au)
Ed Oakley (ed.oakley@southernhealth.org.au)

Version: 6 Date: 13 May 2010

Author's response to reviews:

Dear BMC production team

Re: The comparative rehydration in bronchiolitis study (CRIB)

Journal: BMC Pediatrics

MS ID : 1582445646340298

Formatting changes as below have been made.

Major revision:

Figures 2 and 3 have been re labelled Additional Files 1 and 2. This change has been made in the manuscript, and a section has been added at the end of the manuscript with the title and description of these files.

Please note that I have added myself as an author as the system would not accept the submission with only The research network (PREDICT) as the author.

Yours sincerely

Ed Oakley